Author's response to reviews

Title: Prevalence and severity of antipsychotic related constipation in patients with schizophrenia: a retrospective descriptive study

Authors:

Marc De Hert (marc.de.hert@uc-kortenberg.be)
Liesbeth Dockx (liesbeth.dockx@student.kuleuven.be)
Chiara Bernagie (chiara.bernagie@student.kuleuven.be)
Bie Peuskens (bie.peuskens@uc-kortenberg.be)
Kim Sweers (kim.sweers@uc-kortenberg.be)
Stefan Leucht (Stefan.Leucht@lrz.tu-muenchen.de)
Jan Tack (Jan.Tack@med.kuleuven.be)
Stefan Van de Straete (stefan.van.de.straete@uc-kortenberg.be)
Martien Wampers (martien.wampers@uc-kortenberg.be)
Joseph Peuskens (joseph.peuskens@uc-kortenberg.be)

Version: 3 Date: 21 February 2011

Author's response to reviews:

Dear editor,

We thank you for accepting our paper in your journal. We have made the changes as requested.

Sincerely

Marc De Hert

Reviewer's report

Title: Prevalence and severity of antipsychotic related constipation in patients with schizophrenia: a retrospective descriptive study.

Version: 2 Date: 9 February 2011
Reviewer: Stefan A Mueller-Lissner

Reviewer's report:

Thank you for coming back with the amended manuscript. Though I do still consider the paper of minor impact it has largely been improved, and I will not object publication.

Minor point: it should read "macrogol" not "macrogel".

Changed as requested.

Reviewer's report

Title: Prevalence and severity of antipsychotic related constipation in patients
with schizophrenia: a retrospective descriptive study.

Version: 2  Date: 7 February 2011

Reviewer: Takefumi Suzuki

Reviewer's report:
The reviewer thinks the authors have adequately addressed concerns raised by the referees. In Table 1, brand name for q(Q)uetiapine is missing. Also, "sedative" may appear as "Low-potency typicals". The reviewer has no further comments.

Changed as requested.